Author’s response to reviews

Title: Determinants of Stillbirth among Women Deliveries at Amhara Region, Ethiopia

Authors:

Demeke Lakew (wadela1606@gmail.com)

Dereje Tesfaye (derejetesfaye11@gmail.com)

Haile Mekonnen (hailemekonnen@gmail.com)

Version: 5 Date: 26 Sep 2017

Author’s response to reviews:

Response Letter

A list of modifications and a response to the Editor and reviewers

With this cover letter, we will submit the revised manuscript entitled “Determinants of Stillbirth among Women Deliveries at Amhara Region, Ethiopia (PRCH-D-16-00691R4)” for publication in BMC Pregnancy and Childbirth. We would like to thank for the constructive comments. Based on the comments from the editor, we have made language editing and other corrections of the manuscript, which are indicated below and detailed, are found in the manuscript.

Editor Comments

Thank you for revising your manuscript and reducing the overlap with previous publications. While the overlap has now been sufficiently reduced, the quality of the English and grammar used throughout the manuscript, including the Abstract, does not meet the necessary standards for publication. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service.

• The language of the whole manuscript is edited by English professional. Additionally, we kindly ask that you replace your current "Consent for publication" statement with "Not applicable", since you have not shared any individual participant data.

• The correction is taking in to consideration. Please also ensure that your revised manuscript conforms to the journal style, which can be found in the Instructions for Authors on the journal homepage.

• The final revised manuscript conforms the journal style. I would like to thank the editor for a thorough study of the text, which improved the paper. I hope that now the manuscript is final for publication in BMC pregnancy and childbirth and hence we authors waiting the galley proof and then published after a while. Best regards,

Demeke Lakew Workie

Corresponding author